Author’s response to reviews

Title: Varicocele due to renal arteriovenous malformation mimicking a renal tumor: A case report

Authors:

Peng-chao Li (545084819@qq.com)
Jia-yi Zhang (snake47@yeah.net)
Sheng Liu (Medicaldoc@yeah.net)
Jin-guo Xia (marwen@yeah.net)
Hai-bin Shi (docxuecy@163.com)
Ning-hong Song (drninghong@163.com)

Version: 2 Date: 08 Dec 2017

Author’s response to reviews:

1. At the end of the Introduction there is a need to add a paragraph that explains why this case report is presented (what is unique and adds to the medical knowledge)

Re: Thanks for your professional suggestions. We have added a paragraph to explain what is unique and adds to the medical knowledge (Page 3, line 12 in red color).

2. More information is needed:

Give complete past medical, social, family, and environmental history. What was the patient’s occupation? What medications was the patient on prior to diagnosis? Did the patient smoke, and/or consume alcohol?

Re: Thanks for your professional marking. All above-mentioned were added in the case presentation in red color.

3. Give detailed physical and neurological examination on admission. What was the temperature, pulse, blood pressure and temperature, on admission?
Re: Thanks for your professional marking. All above-mentioned were added in the case presentation in red color.

4. Give all results of laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc)

Re: Thanks for your professional suggestions. Above-mentioned were added in the case presentation and in Table 1.

5. Give information about follow-up for at least 6 months.

Re: Thanks for your professional suggestion. Information about follow-up was added in page 4 in red color.